

PEER REVIEW HISTORY

The peer review history for this manuscript is unavailable due to prepublication legal issues.

Correction: *Diagnostic accuracy study of three alcohol breathalysers marketed for sale to the public*

Ashdown HF, Fleming S, Spencer EA, *et al.* Diagnostic accuracy study of three alcohol breathalysers marketed for sale to the public. *BMJ Open* 2014;**4**:e005811. doi: 10.1136/bmjopen-2014-005811

There is an error in the following sentence:

“One alcohol breathalyser had sensitivity of 26%, corresponding to false reassurance for approximately **one person in four who is** over the limit by the reference standard, at least on the evening of drinking alcohol”.

The part ‘**one person in four who is**’ should read ‘**three people in four who are**’.

Open Access This is an Open Access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: <http://creativecommons.org/licenses/by-nc/4.0/>

© Article author(s) (or their employer(s) unless otherwise stated in the text of the article) 2018. All rights reserved. No commercial use is permitted unless otherwise expressly granted.

BMJ Open 2018;**8**:e005811corr1. doi:10.1136/bmjopen-2014-005811corr1

